Author's response to reviews

Title: Are cervical multifidus muscles active during whiplash and startle? An initial experimental study.

Authors:

Gunter P Siegmund (gunter.siegmund@meaforensic.com)
Jean-Sebastien Blouin (jsblouin@interchange.ubc.ca)
Mark G Carpenter (mark.carpenter@ubc.ca)
John R Brault (john.brault@meaforensic.com)
J Timothy Inglis (tinglis@unixg.ubc.ca)

Version: 6 Date: 4 June 2008

Author's response to reviews:

The following revisions were made to the manuscript:
1) The line numbers were removed.
2) The full stops following the authors' initials were removed.
3) The Figures section was renamed 'Figure legends'.
4) The authors' contribution section was not changed because it already contained a statement that all authors read and approved the final manuscript.
5) Subsection headings were added to the Methods section, and the first subsection heading was set to "Subjects and consent".
6) The figures were cropped to remove unnecessary white space.